Author's response to reviews

Title: Peer education: The effects on knowledge of pregnancy related malaria and preventive practices in women of reproductive age in Edo State, Nigeria.

Authors:

Petra F Mens (p.mens@kit.nl)
Pauline F.D Scheelbeek (p.scheelbeek@kit.nl)
Hind Al Attabi (h.attabi@kit.nl)
Ehijie F.O Enato (enatoefo@uniben.edu)

Version: 4 Date: 29 July 2011

Author's response to reviews:

Dear Editor,

Please find attached the manuscript with the additional changes as requested.

We hope that this is now acceptable for publication.

Kind regards,

Dr. Petra F. Mens